Author's response to reviews

Title: Estimating the incidence of lung cancer attributable to occupational exposure in Iran

Authors:

Alireza Mosavi-Jarrahi (rmosavi@yahoo.com)
Mohammadali Mohagheghi (mamohagheghi@yahoo.com)
Bita Kalaghchi (houmanbita@Hotmail.com)
Yaseman Mousavi-Jarrahi (yasaman_mousaviij@yahoo.com)
Mohammadkazem Noori (mknoori@imidro.org)

Version: 4 Date: 25 March 2009

Author's response to reviews: see over
Author’s response to editor and reviewers' comments:

The following include response to editor request.

Editor's comment:
We would be grateful if you could address further one of the issues that was raised during the initial peer review process: how you dealt with co-exposure to multiple agents. …

Author’s response: The co-exposure and assumptions were elaborated and few references were added to meet the editor’s request.